Author’s response to reviews

Title: Pediatric HIV care and treatment services in Tanzania: implications for survival

Authors:

Mecky Matee (mateemecky@yahoo.com)

Version: 2 Date: 24 Jul 2017

Author’s response to reviews:

Dear Editor,

Responses to your comments:

1) Ethical approval to analyse the data has been clarified.

2) The "Consent for publication" section now read "Not applicable".

3) All authors are included in the submission system

4) Acknowledgements section has been added

5) Figure legends have been removed from the Figures and placed after the References section.

6) If the TB SCREENING & IPT FORM has been removed

7) The response to the reviewers have been removed as well